Author's response to reviews

Title: Alcohol consumption, body mass index and breast cancer risk by hormone receptor status.

Authors:

Aesun Shin (shinaesun@snu.ac.kr)
Sven Sandin (Sven.Sandin@ki.se)
Marie Lof (Marie.Lof@ki.se)
Karen L Margolis (Karen.L.Margolis@HealthPartners.Com)
Kyeezu Kim (kyeezu@gmail.com)
Elisabeth Couto (Elisabeth.Couto@kunnskapssenteret.no)
Hans-Olov Adami (hadami@hsph.harvard.edu)
Elisabete Weiderpass (elisabete.weiderpass.vainio@ki.se)

Version: 2
Date: 12 August 2014

Author's response to reviews:

There are no revisions; this article was first submitted to Breast Cancer Research and rejected, but the editor of that journal suggested that we should submit it to BMC cancer, which we did on the 19th of June 2014 (after we had reformatted the article according to BMC cancer guidelines of course).

We are still waiting for a reply from BMC Cancer, and since it is about 2 months that we submitted the article we kindly ask you to please give us feedback.